Author's response to reviews

Title: Smoking cessation can improve quality of life among COPD patients: Translation and validation of the clinical COPD questionnaire into Greek

Authors:

George Papadopoulos (gnpapa@gmail.com)
Constantine I Vardavas (vardavas@edu.med.uoc.gr)
Maria Limperi (mlimperi@yahoo.com)
Apostolos Linardis (alinardis@ekke.gr)
George Georgoudis (gg@hol.gr)
Panagiotis Behrakis (pbehraki@hsph.harvard.edu)

Version: 4 Date: 8 February 2011

Author's response to reviews:

Dear Editors,

In line with the reviewers request to "see in the results the changes in CCQ total and domains before and after smoking cessation and not only the p value of these changes" we have added one more table, table 4 in the results section with this information included.

Looking forward to your response,

Constantine Vardavas